Reviewer Name: Anonymous

Review Date: 2023-08-06 10:31

Specific Comments To Authors: I would like to see the admission ECG. What were the treatment(s) given? After resolution of LV dysfunction what treatment was the patient kept on?

**Answer:** Thank you for your suggestions. I have attached the admission electrocardiogram. Losartan and diaform treatment was continued during hospitalization. He was advised to continue antidiabetic and antihypertensive drugs after left ventricular function improved.

Reviewer Name: Ajay Kumar Mishra

Review Date: 2023-08-09 00:50

Specific Comments To Authors: In this case reports authors discuss about a patient presenting with TTC following a Hydatid cyst surgery. It is an interesting case report and I have the following recommendations. 1. Provide ECHO image of TTC 2. Did the patient require any ionotropic agent or vasopressor during the surgery? 3. In the first paragraph of discussion discuss about the role of various infections precipitating TTC. Mention the rarity of your case. Refer and cite PMID: 36226557 4. In the second paragraph discuss about the various surgical procedures precipitating TTC 5. Elaborate discussion and cite further articles
Answers: Thank you for your suggestions.

1- We did not record an echo because we initially thought the patient had acute coronary syndrome.

2- The patient did not require inotropic agents and vasopressors during surgery.

3- Mentioned in the manuscript

4- Mentioned in the manuscript

5- Mentioned in the manuscript